Author's response to reviews

Title: A full scale comparative study of methods for generation of functional Dendritic cells for use as cancer vaccines

Authors:

Silvija Jarnjak-Jankovic (silvija.jankovic@rr-research.no)
Hege Hammerstad (Hege.Hammerstad@radiumhospitalet.no)
Stein Saeboe-Larssen (Stein.Saeboe-Larssen@rr-research.no)
Gunnar Kvalheim (Gunnar.Kvalheim@radiumhospitalet.no)
Gustav Gaudernack (gustav.gaudernack@rr-research.no)

Version: 4 Date: 26 June 2007

Author's response to reviews: see over
Reviewer's report
A full scale comparative study of methods for generation of functional Dendritic Title: cells for use as cancer vaccines

Version: 3 Date: 3 June 2007
Reviewer: Marc Dauer
Reviewer's report:

General

Major Compulsory Revisions (that the author must respond to before a decision on publication can be reached)
The authors have sufficiently answered the points made in the initial review.

Minor Essential Revisions (such as missing labels on figures, or the wrong use of a term, which the author can be trusted to correct)

The changes have been made as suggested.

pg. 1, line 15 "were" instead of "was"
pg. 2, line 7 "start" instead of "starts"
pg. 6, line 16 "have" instead of "has"

Discretionary Revisions (which the author can choose to ignore)
None

What next?: Accept after minor essential revisions

Level of interest: An article whose findings are important to those with closely related research interests

Quality of written English: Needs some language corrections before being published

Statistical review: No, the manuscript does not need to be seen by a statistician.

Declaration of competing interests:
I declare that I have no competing interests.